Author’s response to reviews

Title: Health-care seeking behaviour among persons with diabetes in Uganda: an interview study.

Authors:

Katarina Hjelm (katarina.hjelm@lnu.se)
Fortunate Atwine (fortawine@yahoo.com)

Version: 4 Date: 15 September 2011

Author’s response to reviews:

To the Editor of BMC International Health and Human Rights

Dear Editor,

At first we would like to express our sincere gratitude for the most valuable comments given on our manuscript entitled “Health-care seeking behaviour among persons with diabetes in Uganda: an interview-study” Manuscript ID 1293524688507304 submitted to BMC International Health and Human Rights. According to the suggestions changes have been made to the format of our manuscript.

Enclosed, please find a report over changes made.

Sincerely Yours,

Katarina Hjelm, SRNT, MScN, PhD, Professor
School of Health Sciences and Social Work
Linnaeus University, Växjö
S-351 95 Växjö
Sweden

e-mail: katarina.hjelm@lnu.se
Telephone: +46 470 70 83 05, mobile phone: +46 708 31 42 24
Telefax: +46 470 363 10
CHANGES MADE IN THE MANUSCRIPT.

At first we would like to express our sincere gratitude for the most valuable comments given on our manuscript. According to the suggestions the following has been made:

1. Added under Author contributions we have added that "both authors have read and approved the final manuscript.

2. The link to reference number 16 is a functioning link and works.

3. Table 2 is added in the text and the text is checked so that table 1 and 2 matches in the right place in the manuscript.

4. We have deleted information about where tables should be inserted in the text (‘insert here’).